Dynamics of sexual populations structured by a space variable and a phenotypical trait

Sepideh Mirrahimi

CMAP, Ecole Polytechnique, CNRS, INRIA, Route de Saclay, 91128 Palaiseau Cedex, France. mirrahimi@cmap.polytechnique.fr

Gaël Raoul CEFE - UMR 5175, 1919 route de Mende, Montpellier, France gael.raoul@cefe.cnrs.fr

November 20, 2012

Abstract

We study sexual populations structured by a phenotypic trait and a space variable, in a non-homogeneous environment. Departing from an infinitesimal model, we perform an asymptotic limit to derive the system introduced in [25]. We then perform a further simplification to obtain a simple model. Thanks to this simpler equation, we can describe rigorously the dynamics of the population. In particular, we provide an explicit estimate of the invasion speed, or extinction speed of the species. Numerical computations show that this simple model provides a good approximation of the original infinitesimal model, and in particular describes quite well the evolution of the species' range.

Key-words: gene flow, biological invasions, structured populations

1 Introduction

Biological invasion processes play an important role in ecology [18, 34, 46]. Those invasions can for instance be due to the introduction of a new species [22] or to a climate change [20]. Theoretical models, and in particular partial differential models have proven to be useful to study biological invasions as consequences of migration and reproduction [16, 34, 46]. However, those models do not take into account evolutionary processes that can appear during the invasion.

It is known experimentally that biological invasions can involve evolutionary processes which have an impact on the speed, or even on the success of the invasion [27, 19, 45, 17]. For instance, evolutionary changes in dispersal [22, 38, 19, 47, 5] can speed up the invasion process. This is also the case if the environment to invade is inhomogeneous [22, 43, 35, 23]. In this article, we will consider this second case, for a sexual population model.

To study biological invasions in heterogeneous environments, one should consider migration and reproduction, but also mutation and gene flow. It is thus a complicated phenomenon, that has been investigated in numerous articles. In [36], it was shown that the gene flow from the center of the population's range to the edge, can prevent the expansion of the species habitat and be the cause of the limited range of some species. In [25], a model with density regulation was introduced, which showed that a simple environment clime could lead a population to either go extinct, survive in a limited range, or invade the whole environment. This model was later generalized in [9] to several interacting species. In [4], three more detailed models were proposed, describing more accurately the local diversity of the population: A continuum of allelic effect model, a model with two alleles by locus, and an infinitesimal model. The result of this study was surprising, since the various models showed qualitatively different dynamics. In particular, no populations with limited range were obtained with the continuum of allelic effect model, or the model with two alleles by locus, while for the infinitesimal model, a limited range equilibrium and populations invading the whole environment could exist at the same time. Finally, those models have been used successfully to study the effect of global warming on populations' ranges [36, 40, 1, 26].

Most of the studies presented above focus on the cases where limited range is observed. Those cases are indeed of particular interest, since they explain globally the repartition of species. In this article, we will study with equal interest the cases where the population succeeds to invades the whole territory, since we believe it can provide a good description of the dynamics of the population close to the edge of an invasion (even if globally, the range of the population will ultimately remain limited).

In this paper, we first consider an infinitesimal model, where the population is structured by both a phenotypic trait and a space variable. We show that the model of [25] can be obtained from the infinitesimal model through an asymptotic limit (similar to the hydrodynamic limits existing in physics or chemistry [51]). This asymptotic analysis allows us to clarify the derivation of the model, and in particular to highlight the importance of the ratio between the birth rate and the maximal growth rate of the population for the validity of the model of [25]. Then, we simplify the system introduced in [25], thanks to a weak selection limit, to obtain a simple model that is able to describe populations living in a limited range, or populations invading the whole environment. Thanks to that simple model, we are able to describe rigorously the dynamics of the population, and clarify the relation between propagative fronts (invasions or extinctions) and limited range scenario. This simple equation also provides an estimate of the invasion speed of the population, if invasion happens, or of the extinction front if the population dose not succeed to invade. Finally, we illustrate the results with numerical computations, and in particular present comparisons with the dynamics of the initial infinitesimal model.

2 Derivation of simplified models from a structured population model

2.1 The structured population model

We start our analysis from a classical infinitesimal model describing the evolution of a population structured by a phenotypic trait only (see e.g. [8, 14, 32], and [13, 12, 28] for mathematical properties of this kind of models). Let n(t, v) be the population at time $t \ge 0$, with a phenotypic trait $v \in \mathbb{R}$. We assume that the fitness depends on the square of the distance between v and an optimal adaptation trait θ , and is altered by the population size. If we do not take into account the effect of sexual reproduction, the fitness $s[n(t, \cdot)](v)$ of an individual of phenotypic trait v, living among a resident population $n(t, \cdot)$, is given by:

$$s[n(t,\cdot)](v) = r_{max} - \frac{1}{2V_s}(v-\theta)^2 - \frac{r_{max}}{K} \int n(t,w) \, dw, \tag{1}$$

Here $r_{max} > 0$ denotes the maximal growth rate of the population, $\frac{1}{V_s}$ is a measure of the strength of the selection toward the optimal value of the phenotypic trait θ , and K is the carrying capacity of the environment.

The fitness is well-defined in the case of asexual populations: it is the rate of births of offsprings of trait x minus the rate of deaths. In the case of sexual populations however, the situation is more complicated, since reproduction requires two parents, of traits v_* and v'_* , that give birth to an offspring of trait v, usually different from v_* or v'_* . This has two consequences (see [6]):

- We need to define $Q(\cdot, v_*, v'_*)$, the distribution function of the trait of the offspring. We will analyze briefly the properties of Q in Subsection 2.2.
- We have to define the birth rate and the death rate separately. We will assume in this article that the birth rate is a constant, denoted by $\eta > r_{max}$.

In this model, we do not distinguish between males and females (for instance because they have the same distribution) and assume that mating is random and uniform among the population. We also assume that the number of offsprings is proportional to the local population, the idea being that the number of births is proportional to the number of females. Then the evolution of the well-mixed population, structured by a phenotypic trait only, is described by (see [14]):

$$\partial_t n(t,v) = \left[-(\eta - r_{max}) - \frac{1}{2V_s} (v - \theta)^2 - \frac{r_{max}}{K} \int n(t,w) \, dw \right] n(t,v) + \eta \int \int \frac{n(t,v_*)n(t,v'_*)}{\int n(t,w) \, dw} Q(v,v_*,v'_*) \, dv_* \, dv'_*.$$
(2)

Here η is the birth rate ($\eta \ge r_{max}$), that we assume constant within the population (the selection occurs due to the death term).

We next consider populations that are structured by a phenotypic trait v as above, but also by a space variable $x \in \mathbb{R}$. We assume that the selection-mutation process described above occurs locally in space, but that individuals move randomly in space. We model this random movement by a diffusion of rate σ . We additionally assume that the local optimal trait changes linearly in space:

$$\theta(x) = bx. \tag{3}$$

We finally obtain the following model for sexual populations (close to the numerical model used in [15, 1]):

$$\partial_t n(t, x, v) - \frac{\sigma^2}{2} \Delta_x n(t, x, v) = \left[-(\eta - r_{max}) - \frac{1}{2V_s} (v - bx)^2 - \frac{r_{max}}{K} \int n(t, x, w) \, dw \right] n(t, x, v) + \eta \int \int \frac{n(t, x, v_*) n(t, x, v'_*)}{\int n(t, x, w) \, dw} Q(v, v_*, v'_*) \, dv_* \, dv'_*,$$
(4)

where $-\frac{\sigma^2}{2}\Delta_x n(t, x, v)$ models individuals dispersal.

The existence and uniqueness of solutions of (4) has been shown in [41]. Notice that in this article, we do not distinguish the breeding value and the phenotypic trait of an individual, assuming a full heritability of the phenotypic trait, and neglecting the effect of the environment on the phenotypes. In Subsection 6.1 of the Appendix, under the classical assumption that probability law of the phenotypic trait of an individual is a Gaussian centered in its breeding value, we obtain a model similar to (4), where only the constant $-(\eta - r_{max})$ is modified (see e.g. [7]). Our analysis can then be generalized to this more realistic assumption.

2.2 Properties of the sexual reproduction kernel Q

In this subsection, we describe some properties of the sexual reproduction that will be useful in Subsection 2.4. As in e.g. [6, 14], we consider the infinitesimal model, where the trait is affected by an infinite number of loci with infinitesimal and additive effects. For more on the infinitesimal model and discussion of more realistic models, see [50, 49]. If the parents have traits v_* , v'_* , the distribution function $Q(\cdot, v_*, v'_*)$ of the offspring traits is given by the Gaussian distribution

$$Q(v, v_*, v_*') := \frac{1}{\sqrt{\pi V_{LE}}} e^{-\frac{\left(v - \frac{v_* + v_*}{2}\right)^{-1}}{V_{LE}}},$$
(5)

where $V_{LE}/2 > 0$ is the genetic variance of the genotypic values at linkage equilibrium (see [50, 49, 6]), and is assumed to be constant. We consider, in this subsection only, a pure sexual reproduction model, i.e. a well mixed population structured by a phenotypic trait v only, with a constant population size, and without selection:

$$\partial_t n(t,v) = C \left[\int \int \frac{n(t,v_*)n(t,v'_*)}{\int n(t,w) \, dw} Q(v,v_*,v'_*) \, dv_* \, dv'_* - n(t,v) \right],\tag{6}$$

for some constant C > 0. One can check that the total population size $\int n(t, v) dv$ and the mean phenotypic trait of the population $\int v \frac{n(t,v)}{N(t,x)} dv$ are invariant by the equation (6). On the contrary, the second moment of the population phenotypic distribution converges exponentially fast to the trait genetic variance $G^2 = V_{LE}$, and the third moment of the population phenotypic distribution converges to 0: for any initial population (see Theorem 5 in the Appendix),

$$\left| \int \left(v - \int w n(t, w) \, dw \right)^2 \frac{n(t, v)}{\int n(t, w) \, dw} \, dv - G^2 \right|$$

$$\leq e^{-Ct/2} \left| \int \left(v - \int w n(0, w) \, dw \right)^2 \frac{n(0, v)}{\int n(0, w) \, dw} \, dv - G^2 \right|, \tag{7}$$

$$\left| \int \left(v - \int w n(t, w) \, dw \right)^3 \frac{n(t, v)}{\int n(t, w) \, dw} \, dv \right|$$

$$\leq e^{-3Ct/4} \left| \int \left(v - \int w n(0, w) \, dw \right)^3 \frac{n(0, v)}{\int n(0, w) \, dw} \, dv \right|, \tag{8}$$

It has been shown in [6, 50] that (6) has a unique steady-state, that is the Gaussian distribution of variance $G^2 = V_{LE}$. It can indeed be shown (see [29]) that the whole distribution $n(t, \cdot)$ converges exponentially fast to a Gaussian distribution (in the sense of Wasserstein distances, see e.g. [52]).

In Subsection 2.4, estimates (7) and (8) will allow us to close a system of equations describing the evolution of two first moments of the phenotypic distribution of the population $n(t, x, \cdot)$ in (4), and recover the model of [25]. They show that for sexual populations, when C is large (in Subsection 2.4, C will be defined as the ratio between the birth rate and the maximal growth rate of the population), the second and the third moment of the population depend only on Q. They are in particular independent of the fitness landscape and the spatial structure. For asexual populations, the situation would be very different: the phenotypic variance of an asexual population in a mutation-selection equilibrium depends on the fitness landscape (and the mutation model), see e.g. [24, 6, 53, 10].

2.3 Rescaling of the structured population model

Since this term will appear often in this subsection, we write r^* the maximal fitness decreased by the load of the phenotypic variance of the population:

$$r^* := r_{max} - \frac{G^2}{2V_s},$$

where $G^2 = V_{LE}$ is the trait genetic variance.

To simplify (4), we perform the following rescaling introduced in [25] (up to a minor modification):

$$a_1 = r^*, \qquad a_2 = \frac{\sqrt{2r^*}}{\sigma}, \qquad a_3 = (r^*V_s)^{-\frac{1}{2}},$$
$$\widetilde{n}(\widetilde{t}, \widetilde{x}, \widetilde{v}) = \frac{r_{max}}{K} \sqrt{\frac{V_s}{r^*}} n\left(\frac{\widetilde{t}}{a_1}, \frac{\widetilde{x}}{a_2}, \frac{\widetilde{v}}{a_3}\right),$$
$$\widetilde{Q}(v, v_*, v_*') := \frac{1}{a_3} Q\left(\frac{v}{a_3}, \frac{v_*}{a_3}, \frac{v_*'}{a_3}\right).$$

We notice that the rescaled reproduction kernel \tilde{Q} satisfies (7) and (8) with $\tilde{G}^2 = (a_3)^2 G^2 = \frac{G^2}{r^* V_s}$, that is the solution of (6) with \tilde{Q} satisfies:

$$\int \left(v - \int w \, n(t, w) \, dw\right)^2 \frac{n(t, v)}{\int n(t, w) \, dw} \, dv \to \frac{G^2}{r^* V_s} \tag{9}$$

With the above rescaling, (4) becomes, if we omit the tildes:

$$\partial_t n(t, x, v) - \Delta_x n(t, x, v) = -\left[(C - (1 + A/2)) + \frac{1}{2} (v - Bx)^2 + \int n(t, x, w) \, dw \right] n(t, x, v) + C \int \int \frac{n(t, x, v_*) n(t, x, v'_*)}{\int n(t, x, w) \, dw} Q(v, v_*, v'_*) \, dv_* \, dv'_*,$$
(10)

with

$$A := \frac{G^2}{r^* V_s}, \qquad B := \frac{b\sigma}{r^* \sqrt{2V_s}},\tag{11}$$

A	measures the strength of the potential for adaptation to the local optimal trait.					
В	indicates how rapidly the environment changes over space.					
C	measures the ratio between the birth rate and the maximal growth rate of the population.					

Table 1: The interpretation of parameters A, B and C.

$$C:=\frac{\eta}{r^*},$$

and Q satisfying (7) and (8) with $\tilde{G}^2 = A$.

We recall the interpretation of parameters A, B and C in Table 1. Notice that in the case of a population that can only procreate through sexual reproduction, $\eta \ge r_{max}$ and thus $C \in [1, \infty)$. On the contrary, for asexual populations (which we do not consider here), C = 0. The parameter C characterizes indeed the strength of the phenotypic traits mixing caused by the sexual reproduction.

2.4 Derivation of the model (15)

In this subsection, we show that when C is large, the phenotypic variance of the population converges to the constant A, whereas the third moment of the phenotypic distribution of the population converges to 0. The structured population model (10) then reduces to the model (15).

We denote by N, Z, the two first moments of the distribution $n(t, x, \cdot)$:

$$N(t,x) := \int n(t,x,v) \, dv, \quad Z(t,x) := \int v \frac{n(t,x,v)}{N(t,x)} \, dv,$$

Then (see Theorem 2 in Subsection 6.3 of the Appendix), N and Z satisfy the following unclosed equations:

$$\partial_t N(t,x) - \Delta_x N(t,x) = \left[1 + \frac{1}{2} \left(A - \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} \, dv \right) - \frac{1}{2} (Z(t,x) - Bx)^2 - N(t,x) \right] N(t,x).$$
(12)

$$\partial_t Z(t,x) - \Delta_x Z(t,x) = 2\partial_x (\log N(t,x)) \partial_x Z(t,x) + (Bx - Z(t,x)) \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} dv - \int (v - Z)^3 \frac{n(t,x,v)}{N(t,x)} dv.$$
(13)

The term $2\partial_x(\log N)\partial_x Z$ is referred to as the swamping, or asymmetric gene flow term (see [31] and [36, 25, 40, 42]). This term models the fact that the mean phenotype of low density areas are greatly influenced by the phenotypes of neighboring high density areas. It is interesting to notice that this term does not come from the sexual reproduction term, but from the diffusion term:

$$\Delta_x Z(t,x) = \int v \frac{\Delta_x n(t,x,v)}{N(t,x)} \, dv - \frac{Z(t,x)}{N(t,x)} \Delta_x N(t,x) + 2\partial_x (\log N(t,x)) \partial_x Z(t,x).$$

Equations (12) and (13) can also be obtained for asexual populations. However, the moment closure that we will present next, and that will allow us to derive the model of [25] from (12) and (13) does not apply to asexual populations. This is fortunate, since the cases where the population survives in a limited range, obtained in [25] only make sense for sexual populations.

To close the equations (12) and (13), we notice that (10) can be written:

$$\partial_t n(t,x,v) - \Delta_x n(t,x,v) = C \left[\int \int \frac{n(t,x,v_*)n(t,x,v'_*)}{\int n(t,x,w) \, dw} Q(v,v_*,v'_*) \, dv_* \, dv'_* - n(t,x,v) \right] \\ + \left[1 + \frac{A}{2} - \frac{1}{2}(v - Bx)^2 - \int n(t,x,w) \, dw \right] n(t,x,v)$$

Therefore if C is very large (that is $C >> \max\{1, A, B\}$), the first term dominates the dynamics of the population. This first term corresponds to the "pure" sexual reproduction equation (6). It does not affect the total population N(t, x) or the mean phenotypic trait Z(t, x), but it acts on the second and the third moments of the phenotypic distribution of the population, as described in Subsection 2.2. The recalled reproduction kernel \tilde{Q} (denoted by Q here) satisfies (7) and (8) with $\tilde{G}^2 = \frac{G^2}{r^*V_s} = A$ (see (9)). If C is large, it is then natural to assume that at all time t > 0and all locations $x \in \mathbb{R}$,

$$\int (v - Z(t, x))^2 \frac{n(t, x, v)}{N(t, x)} dv \sim A, \qquad \int (v - Z(t, x))^3 \frac{n(t, x, v)}{N(t, x)} dv \sim 0.$$
(14)

We can then use the estimates (14) to close the system of equations (12) and (13) and to obtain the model (15) of [25]:

$$\begin{cases} \partial_t N(t,x) - \Delta_x N(t,x) = \left(1 - \frac{1}{2}(Z(t,x) - Bx)^2 - N(t,x)\right) N(t,x), \\ \partial_t Z(t,x) - \Delta_x Z(t,x) = 2\partial_x (\log N(t,x)) \partial_x Z(t,x) + A(Bx - Z(t,x)), \end{cases}$$
(15)

Our derivation shows that the model (15) is valid (in the sense that it is the formal asymptotic limit of the infinitesimal (4)) if:

- The reproduction is sexual,
- The reproduction kernel satisfies (7) and (8),
- C is large.

With the asymptotic that we consider in this article (C large), the phenotypic variance of the population depends only on the phenotypic variance of the offsprings ($G^2 = V_{LE}$, see Subsection 2.3), and neglects the effect of migration on the phenotypic variance. If $C \ge 1$ is small, numerical computations in Subsection 4.3 show that this simplification may not be very accurate, away from the edge of the propagating front.

Unfortunately, there is no existence and uniqueness theory for (15). The difficulty comes from the term $\partial_x(\log N(t,x)) = \frac{\partial_x N(t,x)}{N(t,x)}$, which is not defined when N is not bounded away from 0. One possibility to avoid this difficulty is to multiply the second equation of (15) by N. But even in this case, the equation on Z is not defined when N(t,x) = 0. This problem will persist in the simplified model that we will derive in the next subsection, where we will discuss the problem of the definition of solutions in more details (see Example 1, and Subsection 6.4 in the Appendix).

2.5 Derivation of the model (18)

In this subsection, we show that when A and B are small, the model (15) simplifies further. To do so, we first apply the following change of variable:

$$\check{N}(\check{t},\check{x}) := N\left(\frac{\check{t}}{A},\frac{\check{x}}{\sqrt{A}}\right), \qquad \check{Z}(\check{t},\check{x}) := \frac{1}{\sqrt{2}}Z\left(\frac{\check{t}}{A},\frac{\check{x}}{\sqrt{A}}\right).$$

To simplify the notations, we omit the breves in the new unknown functions. The rescaled model becomes

$$\begin{cases} \partial_t N(t,x) - \Delta_x N(t,x) = \frac{1}{A} \left[1 - (Z(t,x) - Dx)^2 - N(t,x) \right] N(t,x), \\ \partial_t Z(t,x) - \Delta_x Z(t,x) = 2\partial_x (\log N(t,x)) \partial_x Z(t,x) + (Dx - Z(t,x)), \end{cases}$$
(16)

where $D = \frac{B}{\sqrt{2A}}$. Now, if we assume that A and B are small, while D remains constant, N and Z are related by the simple relation:

$$N(t,x) \sim 1 - (Z(t,x) - Dx)^2.$$
(17)

Notice that $D = \frac{b\sigma}{2\sqrt{r^*G}}$. If $\frac{1}{V_s} \to 0$, then $A, B \to 0$ while D remains constant. The model (18) can then be seen as a weak selection limit of (15). This is however not the only case where this model applies: it would also apply, for instance, if $G = \sigma \to 0$. In [25], the range of A that has been considered is $A \in [0.001, 1]$, and the range of B when the population does not go extinct is $B \in [0.01, 1]$. Thus our approximation, assuming that A and $B \sim \sqrt{A}$ are small, is coherent with [25]. As we will see in Section 3, this model will provide relevant approximations of the dynamics of the population, even when A is not very small.

Then, we obtain the simpler model (18), on Z only:

$$\partial_t Z(t,x) - \Delta_x Z(t,x) = -4 \frac{(\partial_x Z(t,x) - D)(Z(t,x) - Dx)}{1 - (Z(t,x) - Dx)^2} \partial_x Z(t,x) + (Dx - Z(t,x)).$$
(18)

Notice that $N \ge 0$ implies that this approximation makes sense only if $|Z(t,x) - Dx| \le 1$. Therefore we consider only solutions of (18) such that $Z(t,x) \in [Dx - 1, Dx + 1]$.

As we can see in (16), the limit that we performed corresponds to an infinitely fast reaction term. This model is thus inadequate to capture the extinction phenomenon that occurs at the infinitesimal level (see Section 4). This simple model is however able to capture the two other possible behaviors of the population: propagation to the entire environment and survival in a limited range.

Remark 1 Another simplification had been proposed in [25], where the equation on N was replaced by

$$N := k \exp\left(\beta \left(1 - A(Z - Bx)^2\right)\right). \tag{19}$$

With this ansatz, the model (15) simplifies considerably:

$$\partial_t Z(t,x) - \Delta_x Z(t,x) = A(Bx - Z(t,x)) \left[1 - 4\beta \partial_x Z(t,x)(B - \partial_x Z(t,x))\right]$$

However, the simplification (19) seems independent of (15). Our simplification has the advantage to rely on a clearer assumption: (18) is the formal limit of (15) when A is small.

The equation (18) has a singularity for $Z = Dx \pm 1$. The existence of singularities is an obstacle to have a well-defined problem. However, as we will see in Subsection 3.1, the singularities are fundamental to produce propagating fronts. We have not been able to find a good notion of solution, that is a definition of solutions that would imply both the existence of solution for general initial data, and the uniqueness of those solutions. Nevertheless we are able to show in Subsection 6.4 of the Appendix (Theorem 3) that (18) admits viscosity solutions (see Subsection 6.4 in the Appendix for the definition of viscosity solutions and [11, 2] for general introduction to the theory of viscosity solutions). Unfortunately the viscosity criterion is not enough to select a unique solution. We give a counter-example below:

Example 1 Non-uniqueness for equation (18): We have the two following solutions to equation (18):

$$Z_1(t,x) = Dx - 1, \quad \text{for all } (t,x) \in \mathbb{R}_+ \times \mathbb{R},$$
$$Z_2(t,x) = Dx - e^{-t}, \quad \text{for all } (t,x) \in \mathbb{R}_+ \times \mathbb{R}.$$

The first solution corresponds to N(t,x) = 0, while the second corresponds to $N(t,x) = 1-e^{-2t}$, that is to a the spontaneous apparition of a population. The biologically relevant solution is thus first one: if initially Z(0,x) = Dx - 1 for all $x \in \mathbb{R}$, we expect that $W(t, \cdot) \equiv -1$, for all $t \in \mathbb{R}_+$. However, an existence theory able to select the biologically relevant solution for any initial population doesn't exist yet (see also Subsection 6.4 in the Appendix).

3 Dynamics of the population, and propagating fronts

3.1 Theoretical study of the simplified model (18)

In this subsection, we study the existence of propagating fronts and steady states for equation (18). Since the optimal trait at the point x of space is Dx, we expect a propagating front Z, with Z a solution of (18), to be of the following form

$$Z(t,x) = Dx + U(x - \nu t).$$

Using the above notation, the equation (18) becomes

$$-\nu U' - U'' = -4\frac{U'U}{1 - U^2}(U' + D) - U,$$

with $U : \mathbb{R} \mapsto [-1, 1]$. If we denote by V := U' the derivative of U, finding a propagating front is equivalent to finding a solution defined on \mathbb{R} to the ODE

$$\frac{d}{dt}(U,V)(t) = \left(F_U(U,V), F_V(U,V)\right),$$

where the vector field (F_U, F_V) is given by:

$$\begin{cases} F_U(U,V) = V, \\ F_V(U,V) = -\nu V + 4\frac{UV}{1-U^2}(V+D) + U. \end{cases}$$
(20)

Using this new formulation of the problem, we show in Subsection 6.5 of the Appendix (Theorem 5) that for any D > 0, there exists a minimal speed $\nu_D \in \mathbb{R}$ such that (18) has a propagating front of speed ν , $Z(t, x) = Dx + U(x - \nu t)$, satisfying

$$U(x) \to 0 \text{ as } x \to -\infty, \quad U(x) \to -1 \text{ as } x \to +\infty,$$

if and only if $\nu \geq \nu_D$. Just as for classical invasion models (like the Fisher-KPP model, see [34, 46]), only the front of minimal speed is biologically relevant: the fronts of speed $\nu > \nu_D$ can be obtained only with very particular initial populations. One can then consider that the propagation speed of the population is ν_D . The corresponding propagating front has a sharp edge: there exists $x(t) = x_0 + \nu_D t$ such that $N(t, x) = 1 - (Bx - Z(t, x))^2 \neq 0$ for x < x(t), and N(t, x) = 0 for $x \geq x(t)$. Moreover, $\partial_x Z(t, x) \to 0$ as $x \to x(t)$, x < x(t), that is the mean phenotypic trait of the population is almost constant close to the edge of the population's range. We represent some propagating fronts in Figure 1.

In Subsection 6.5 of the Appendix, we show that (18) is also able to describe the situations where the population survives in a limited range, but does not succeed to invade the whole space. More precisely, we show that (18) has a non-trivial steady-state (that is the population has a limited range) if and only if $\nu_D < 0$ (that corresponds to the case where the propagating front described above is indeed an extinction front). Those steady-states Z(t,x) = Dx + U(x)satisfy

$$U(x) \to 1 \text{ as } x \to -\infty, \quad U(x) \to -1 \text{ as } x \to +\infty.$$

Similarly to the propagating fronts that we described above, provided that $\nu_D < 0$, there exists a whole family of steady-states (parameterized by $V(0) = U'(0) \in [-K_D, 0)$), only one of which being biologically relevant (the one satisfying $V(0) = U'(0) = -K_D$). Those steady states also have sharp edges: there exist x_{min} , x_{max} such that the population is present on the interval $[x_{min}, x_{max}]$ only, and close to the edges of the population's range, the mean phenotypic trait of the population is almost constant. We represent some steady populations in Figure 2. As we can see numerically in Figure 2 (or theoretically thanks to (20)), the mean phenotypic trait of the population is almost constant as soon as D is large $(D \gtrsim 4)$.

3.2 Estimation of the speed of the propagating fronts

The correspondence between propagating fronts of (18) and solutions of (20) provides a simple way to determine numerically the minimal speed and shape of the propagating fronts of (18). As we show in Subsection 6.6 of the Appendix, it also provides a way to estimate analytically the speed of the propagating fronts: for D > 0,

$$\nu_D \sim \frac{1}{20} \left(7 \left(\frac{2}{D} - 9D \right) + 3 \sqrt{\left(\frac{2}{D} - 9D \right)^2 + 40} \right).$$
(21)

In Figure 9, we see that this analytical approximation of ν_D is remarkably close to the values of ν_D obtained numerically. If we come back to the scaling of (15) or (10), the population described by n or (N, Z) would then propagate at a speed $\nu_{A,B} \sim \frac{1}{\sqrt{A}} \nu_{B/\sqrt{2A}}$.

We notice that (21) provides an estimate $D_{crit} \sim \frac{2}{3}$ for the critical value of D such that $\nu_D = 0$, which corresponds to the limit between cases where the species invade the whole



Figure 1: Propagating fronts. We represent the mean phenotypic trait $x \mapsto Z(x - \nu_D t)$ and the corresponding population size $x \mapsto N(x - \nu_D t)$, for D = 0.1 in the first line and D = 5 in the second. For D = 0.1, the front is an invasion front, $\nu_{0.1} \sim 9.7 > 0$, whereas for D = 5, it is a depopulation front, $\nu_5 \sim -9.25 < 0$. On the graphs representing the mean phenotypic traits, we also represent the optimal trait Z = Dx (in black), and the two lines $Z - Dx = \pm 1$ (in blue), which correspond to the limits of the domain where Z has a meaning (we recall that $Z(t,x) \in [Dx - 1, Dx + 1]$).

territory $(D < D_{crit})$, or remain in a limited range $(D > D_{crit})$. Notice that this estimation of the critical value is different from the one obtained in [25] where $D_{crit} \sim 1$ was derived from the approximation that $N \ll 1$. It seems that this approximation does not apply in the weak selection limit considered here (see Subsection 2.5).

We can also derive from the above estimate an approximation of the shape of the propagating fronts:

$$Dx + U_{\nu_D}(x) \sim Dx - \left(\left(1 - \frac{\beta}{D+\beta} \right) e^{-\frac{D+\beta}{2}x} + \frac{\beta}{D+\beta} \right)^{-2}, \quad x \in \mathbb{R}_-,$$

where $\beta = \frac{1}{2} \left(\sqrt{\nu_D^2 + 4} - \nu_D \right) - D$, and $Dx + U_{\nu_D}(x) = Dx - 1$ on \mathbb{R}_+ . Finally, N_{ν_D} can be estimated thanks to the formula $N_{\nu_D}(x) = 1 - U_{\nu_D}^2$.



Figure 2: Population at equilibrium with a limited range. We represent the mean phenotypic trait Z(x) and the corresponding population size N(x), for D = 0.7 in the first line and D = 5 in the second. On the graphs representing the mean phenotypic traits, we also represent the optimal trait Z = Dx (in black), and the two lines $Z - Dx = \pm 1$ (in blue), which correspond to the limits of the domain where Z has a meaning (we recall that $Z(t, x) \in [Dx - 1, Dx + 1]$).

4 Numerics

In this section, we analyze numerically the dynamics of the structured population model (10), and compare it with the propagating fronts and steady-states obtained from the simplified model (18).

4.1 Numerical scheme

The difficulty for the numerical resolution of (10) is that a direct discretization of the birth term leads to a very slow algorithm (of order k^3 , where k is the number of nodes in the v variable). Fortunately, with the classical sexual reproduction kernel (5), we can write the reproduction term as a double convolution, as noticed in [50]:

$$\int \int \frac{n(t,x,v_*)n(t,x,v'_*)}{\int n(t,x,w)\,dw} Q(v,v_*,v'_*)\,dv_*\,dv'_* = \frac{1}{\int n(t,x,w)\,dw} \left(\tilde{Q} *_v f(t,x,\cdot) *_v f(t,x,\cdot)\right)(2v),$$



Figure 3: Comparison of the minimal speed ν_D of propagating waves of the simplified model (18) (in blue), with the approximated speed (21) (in green).

where Q is defined by (5) with $\gamma^2 = A$, and $\tilde{Q}(v) := \frac{1}{A\sqrt{2\pi}}e^{-\frac{v^2}{8A^2}}$. This formula allows us, as described in [50] to use a spectral method to compute the reproduction term, and the computational cost of the birth term then becomes of the order of a Fast Fourier Transform, that is $k \log k$. Coupled to a classical finite difference scheme for the rest of the equation, this produces an algorithm reasonably fast.

For the simplified model (18), we can use the phase plane (U, V) described in Subsection 3.1 to compute numerically the steady-states and propagating fronts of (18). We can in particular estimate numerically ν_D (see Theorem 5), and obtain a numerical approximation of the corresponding front.

4.2 Dynamics of the population

By performing a numerical resolution of (10), we can observe the three possible behaviors described in [25]: For given A and C, we observe extinction if B is large (see Figure 4), convergence to a steady-state for intermediate values of B (see Figure 5), and propagation if B is small (see Figure 6).

For C = 20, Figure 7 describes the dynamics of populations depending on A and B. The dots come from the numerical resolution of (10) and they determine the limit between the zones where populations propagate, remain in a limited range or get extinct. We can also use the simplified model (18) to determine the border between the zones where populations succeed to propagate (if $D = \frac{B}{\sqrt{2A}} < D_{crit} \sim \frac{2}{3}$), and where populations remain in a limited range (if $D = \frac{B}{\sqrt{2A}} > D_{crit} \sim \frac{2}{3}$). This border is represented by a continuous line in Figure 7. We notice that the agreement between the theoretical border and the numerical results is good.

As explained in Subsection 2.5, the simplified model (18) is unable to describe the extinction phenomena. However, we can still obtain formally the border between the zones of extinction, and limited range. Indeed, numerically we observe that when a steady population is small, it tends to have a constant mean trait, that is Z is constant. Using this purely formal ansatz, we can use (15) to get a criterion for extinction or survival of populations: it depends on the sign



Figure 4: Extinction for A = 0.01, B = 1.3, C = 20. The graph represents the total population $x \mapsto N = \int n(t, x, v) dv$ for four successive times.



Figure 5: Limited range, for A = 0.01, B = 0.3, C = 20. The three graphs represent respectively the population $(x, v) \mapsto n(t, x, v)$, the total population at each location $x \mapsto N = \int n(t, x, v) dv$, and the mean phenotypic trait of the population, $x \mapsto Z = \int v \frac{n(t, x, v)}{N(t, x)} dv$.



Figure 6: Propagation for A = 0.01, B = 0.1, C = 20. The three graphs represent respectively the population $(x, v) \mapsto n(t, x, v)$, the total population at each location $x \mapsto N(t, x) = \int n(t, x, v) dv$, and the mean phenotypic trait of the population, $x \mapsto Z(t, x) = \int v \frac{n(t, x, v)}{N(t, x)} dv$.

of the principal eigenvalue of the linear operator

$$\mathcal{L}(N)(x) = \Delta N(x) + \left(1 - \frac{B^2}{2}x^2\right)N(x),$$

that is the sign of $1 - \frac{B}{\sqrt{2}}$ (the corresponding eigenfunction being $N(x) = e^{-\frac{Bx^2}{2\sqrt{2}}}$). Following this formal argument, extinction would occur if $B > \sqrt{2}$. In Figure 7 we also represent the line $B = \sqrt{2}$, that appears to be close to the border obtained by numerical resolution of (10).

4.3 Comparison between the structured population model (10) and the simplified model (18)

In Section 2, we showed that the structured population model (10) converges formally to the simplified model (18) when C is large and A, B are small. Then, in Subsection 3.1, we have described the steady-states and traveling waves of the simplified model (18) using a phase plane analysis. In Figure (8), we compare the numerical resolution of the structured population model (10) with the propagating fronts of the simplified model (18) for A = 0.01, B = 0.1 and C = 20.



Figure 7: Dynamics of the structured population model (10). The dots come from the numerical resolution of (10) with C = 20. The continuous lines come from the approximations presented in Subsection 4.2.



Figure 8: Comparison of propagating wave profiles of the structured population model (10) with the propagating wave of minimal speed ν_D for the simplified model (18). We depict the total population N for both models (left) and the mean phenotypic trait Z for both models (right), using the following parameters: A = 0.01, B = 0.1, and C = 20.

We can compare numerically the propagation speed for models (15) and (18) for A = 0.01 small. As we can see in Figure 9, the speed ν_D obtained from the simplified model seems a good approximation of the speed of propagating fronts from the structured population model (10) when C is large (C = 20 on figure 9). When C is small (C = 1, which is the minimal possible value of C for a sexual population), the approximation is much less accurate. The assumption that C is large, which we used to derive the model (15) from (4) in Subsection 2.4 thus seems necessary for the simpler models (15) and (18) to be relevant.

To investigate further the influence of C on the properties of the population, we consider the case where A = 0.01, B = 0.5 and C = 1 or C = 20. We observe in Figure 10 that the values of the total population $N(t, x) = \int n(t, x, v) dv$ and the mean phenotypic trait $Z(t, x) = \int v \frac{n(t, x, v)}{N(t, x)} dv$ for the propagating fronts obtained numerically for C = 1 and C = 20 are close. However, in the case C = 1, the value of the phenotypic variance of the population is far from the value $\int (v - Z(t, x))^2 \frac{n(t, x, v)}{N(t, x)} dv = A$ used in our asymptotic analysis (see Subsection 2.4). This is because in our asymptotic analysis (Subsection 2.4), we have neglected the phenotypic variance produced by the migration, which probably explains the dependence of the population's propagation speed in the coefficient C.

Notice however that close to the edge of the propagating front, the assumption that the phenotypic variance of the population is A seems accurate. This is probably due to the fact that in this region, the mean phenotype Z is almost constant in x (see Subsection 3.1), migration has then little effect on the phenotypic variance of the population. Our approximation that C is large thus fails to reproduce the phenotypic variance of a population far from the edge of the habitat (for $C \ge 1$ small), but describes it quite well at the edge of the habitat (even for C = 1, see Fig. 10).



Figure 9: Comparison of the propagating wave speed of the structured population model (10) for C = 1 (in black), C = 20 (in red) and the minimal speed ν_D of propagating waves of the simplified model (18) (in blue).

5 Discussion

The study of populations living in heterogeneous environments is challenging, since many biological processes are combined: migration, mutation, gene flow... Understanding the dynamics of such populations is however crucial to study the repartition area of species, or the effect of global warming on a species' range. Several models have been proposed to study such populations.

In [36], a model describing the evolution of the mean phenotypic trait and the size of the population was introduced (the model also allowed the optimal phenotypic trait to evolve in time). This model has an exact solution, which is very useful in practice (see e.g. [26]). The model however does not allow for any density dependence in the fitness function, which is problematic, since the population's size can grow indefinitely. The absence of density regulation leads indeed to a strong gene flow from the center of the habitat to the edge, which prevents any species to invade the whole environment.

A density dependence was then introduced in [25], which leads to a model able to capture three possible outcomes: the population can either go extinct, survive in a limited area, or spread to



Figure 10: Comparison of propagating wave profiles of the structured population model (10) for C = 1 (in red) and C = 20 (in blue). The other parameters are A = 0.01, B = 0.5, and the speeds of propagation are respectively 0.237 and 0.792. We depict the total population N (left), the mean phenotypic trait Z (right), and the phenotypic variance (bottom) of the population. Notice the different scales for the phenotypic trait axis in those two last plots. The variance of the population when C is very large is $G^2 = A = 0.01$.

the entire environment. No exact solution is known for this more realistic model (except for a population that is perfectly adapted everywhere), and simplifications are then necessary to understand its dynamics.

In [25], the phenotypic variance of the population was assumed constant, which may not be realistic. More detailed models where thus introduced in [4]: a continuum of alleles model, a model for two alleles per locus, and an infinitesimal model. Surprisingly, the conclusion was then very different from [25]: no stationary populations were observed for the two first models, whereas both convergence to a steady population with limited range or invasion to the whole space were possible for the infinitesimal model. In Figure 10, numerical resolutions of the infinitesimal model suggest that the phenotypic variance in the population is minimal at the edge of the population range. This result is quite different from the results of [4], where it is explained that gene flow can inflate the genetic variance at the edge of the population's range, enabling its propagation. In this article, we have shown that the model of [25] can be obtained from the infinitesimal model through an asymptotic limit (see Subsection 2.4). This asymptotic analysis shows an explicit link between those two models, which enables us to compare them numerically. Numerically, we observe no qualitative difference between the dynamics of the infinitesimal model and the model of [25]. In particular, the existence of steady populations seems equivalent to the existence of extinction fronts, just as for the simplified model (18). Our analysis emphasizes the importance of the parameter C (the birth rate divided by the maximal growth rate of the population): the model of [25] is the asymptotic limit of the infinitesimal model when C is large. Notice that the cases where $C \leq 1$ may be interesting, since it allows the investigation of populations that do not only reproduce sexually.

In our asymptotic analysis, we have assumed (as a consequence of our asymptotic limit C large) that the phenotypic variance of the population is constant, and is not affected by migration. As we showed, this assumption is quite rough when C is small, but it remains accurate close to the edge of the invasion front for any $C \ge 1$. This result is quite different from the result obtained in [4] with the continuum of alleles model and the model for two alleles per locus. It was observed there that close to the edge of the species' range, mutations increase the phenotypic variance of the population ultimately allowing the population to invade the whole environment under any circumstance. A natural continuation of our work would be to modify our asymptotic limit to allow the phenotypic variance of the population to evolve. Numerically, an approximation of the type $\int (v - Z(t, x)^2 \frac{n(t,x,v)}{N(t,x)} dv \sim A + \varepsilon(A, B, C) \partial_x Z(t, x)^2$ (where $\varepsilon(A, B, C) \to 0$ as $C \to 0$) seems to be very close to the numerical resolution of the infinitesimal model. This would be the next step to understand the connections between the infinitesimal model and the continuum of alleles model proposed in [4].

From the model of [25], we have derived the simplified model (18). This new model can be seen as a weak selection limit of [25], and is interesting because its simplicity allows a good understanding of its dynamics, as well as a good approximation of the invasion speed or extinction speed. If a steady population with limited range exists (for a given set of parameters of the equation (4)), we have shown using the simplified model (18) that depopulation fronts exist. A perfectly adapted population occupying the entire space may be stable, but as soon as the population has a bounded range (N(0, x) = 0 outside a bounded interval, that is Z(0, x) = Dx+1when x < -C, Z(0, x) = Dx - 1 when x > C for some C > 0), a depopulation front will shrink that range, and finally the population will converge to the limited range steady population.

Following [36], the models proposed in [36, 25, 4] are mostly used in the case of limited range. We believe however that those models can also be interesting to study invasion or extinction processes, as an extension of classical invasion models (see [16, 34, 46]) including evolutionary effects. In particular, the mathematical tools presented here may be useful to study the evolution of species' range under the effect of global warming. The idea would not be to model the whole population, but simply a part of it, close to an edge of its range. In the context of global warming (where this type of model have been widely used in the case of limited range [1, 26]) those models could be used to understand how fast the species will invade towards the north (if we consider a population living in the northern hemisphere), or how fast it will go extinct on the southern edge of its range. Our explicit approximation of those speeds, provided in Subsection 3.2 can then be useful (after a modification to include the effect of global warming, that we did not consider here).

As noticed in Subsections 2.4 and 2.5, a good definition of solutions of (15) and (18) is still

missing, when the population N(t, x) is not initially bounded away from 0. This case is of particular importance, since those models can be used to study invasions of new environments. This theoretical problem has the concrete implication that classical finite difference schemes, for (18) at least, fail to select the biologically relevant solutions of the model (see Example 1). An investigation of an existence and uniqueness theory would then be interesting. It seems that the approximation scheme (25) (or a similar approximation for (15)) could be a good intermediate step.

Acknowledgments: SM benefits from a 2 year "Fondation Mathématique Jacques Hadamard" (FMJH) postdoc scholarship. She would like to thank Ecole Polytechnique for its hospitality. GR has been supported by Award No. KUK-I1-007-43 of Peter A. Markowich, made by King Abdullah University of Science and Technology (KAUST). The authors would like to thank Laurent Desvillettes for introducing the problem to them. The authors are ordered alphabetically.

6 Appendix

6.1 Case of a population structured by a Breeding value

The phenotypic trait is produced by a combination of the breeding trait, determined by the genotype of the individual, and environmental factors. A common assumption is to consider that the probability distribution of the phenotype v of an individual is given by a Gaussian distribution centered around its breeding value p, with variance V_E . The fitness of a population of breeding value p is then given by the average fitness of the Gaussian distribution. If n(t, x, p) is a population at time $t \geq 0$ structured by a space variable x and a breeding value p, (4) then becomes:

$$\partial_t n(t,x,p) - \frac{\sigma^2}{2} \Delta_x n(t,x,p) \\
= \left[-\left(\eta + \frac{1}{2V_s} V_E - r_{max}\right) - \frac{1}{2V_s} (p - bx)^2 - \frac{r_{max}}{K} \int n(t,x,q) \, dq \right] n(t,x,p) \\
+ \eta \int \int \frac{n(t,x,p_*)n(t,x,p'_*)}{\int n(t,x,q) \, dq} Q(p,p_*,p'_*) \, dp_* \, dp'_*,$$

since

$$\int -\frac{1}{2V_s} (v - bx)^2 \frac{e^{\frac{-(v-p)^2}{2V_E}}}{\sqrt{2\pi V_E}} \, dv = -\frac{1}{2V_s} (p - bx)^2 - \frac{1}{2V_s} V_E.$$

6.2 Evolution of the moments of the population density for the pure selection model (6)

Theorem 1 If Q is defined by (5), any solution of (6) satisfies (7) and (8).

Proof of Theorem 1:

We assume, without loss of generality, that $\int n(0, v) dv = 1$, $\int v n(0, v) = 0$, and C = 1 (all cases can be deduced through a change of variables). Then for all times $t \ge 0$,

$$\int v \, n(t,v) \, dv = 0$$

We multiply equation (6) by v^2 and integrate to obtain

$$\begin{aligned} \frac{d}{dt} \int v^2 n(t,v) \, dv &= \int \int \left(\int v^2 Q(v,v_*,v_*') \, dv \right) n(t,v_*') n(t,v_*) \, dv_* \, dv_*' - \int v^2 n(t,v) dv \\ &= \int \int \left(\gamma^2 + \left(\frac{v_* + v_*'}{2}\right)^2 \right) n(t,v_*') n(t,v_*) \, dv_* \, dv_*' - \int v^2 n(t,v) dv \\ &= \gamma^2 - \frac{1}{2} \int v^2 n(t,v) dv. \end{aligned}$$

We deduce that the variance of n converges exponentially fast to $G^2 = 2\gamma^2$ for any initial condition n^0 :

$$\left|\int v^2 n(t,v)\,dv - 2\gamma^2\right| \le e^{-t/2} \left|\int v^2 n(0,v)\,dv - 2\gamma^2\right|.$$

Similarly we have

$$\begin{aligned} \frac{d}{dt} \int v^3 n(t,v) \, dv &= \int \int \left(\int v^3 Q(v,v_*,v_*') \, dv \right) n(t,v_*) n(t,v_*') \, dv_* \, dv_*' - \int v^3 n(t,v) dv \\ &= \int \int \left(\frac{v_* + v_*'}{2} \right)^3 n(t,v_*) n(t,v_*') \, dv_* \, dv_*' - \int v^3 n(t,v) dv \\ &= -\frac{3}{4} \int v^3 n(t,v) dv. \end{aligned}$$

It follows that $\left|\int v^3 n(t,v) \, dv\right| \le e^{-3t/4} \left|\int v^3 n(0,v) \, dv\right|.$

6.3 Unclosed equations on the moments of the population distribution moments

Theorem 2 If n is a solution of (10), then the moments of the phenotypic distribution of n satisfy:

$$\partial_t N(t,x) - \Delta_x N(t,x) = \left[1 + \frac{1}{2} \left(A - \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} \, dv \right) - \frac{1}{2} (Z(t,x) - Bx)^2 - N(t,x) \right] N(t,x).$$
(22)

$$\partial_t Z(t,x) - \Delta_x Z(t,x) = 2\partial_x (\log N(t,x))\partial_x Z(t,x) + (Bx - Z(t,x)) \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} dv - \frac{1}{2} \int (v - Z)^3 \frac{n(t,x,v)}{N(t,x)} dv.$$
(23)

Proof of Theorem 2: To obtain (22), we integrate (10) along v:

$$\partial_t N(t,x) - \Delta_x N(t,x) = \int \partial_t n(t,x,v) - \Delta_x n(t,x,v) \, dv$$

= $\left[1 + \frac{A}{2} - N(t,x) \right] N(t,x) - \frac{1}{2} \int (v - Bx)^2 n(t,x,v) \, dv$
= $\left[1 + \frac{1}{2} \left(A - \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} \, dv \right) - \frac{1}{2} (Z(t,x) - Bx)^2 - N(t,x) \right] N(t,x).$

The second equation, (23), is obtained as follows:

$$\partial_t Z(t,x) - \Delta_x Z(t,x) = \partial_t \int v \frac{n(t,x,v)}{N(t,x)} \, dv - \Delta_x \int v \frac{n(t,x,v)}{N(t,x)} \, dv$$
$$= \int \frac{v}{N(t,x)} \left(\partial_t n(t,x,v) - \Delta_x n(t,x,v) \right) \, dv$$
$$- \left(\partial_t N(t,x) - \Delta_x N(t,x) \right) \frac{Z(t,x)}{N(t,x)}$$
$$+ 2\partial_x (\log N(t,x)) \partial_x Z(t,x),$$

and thus,

$$\begin{split} \partial_t Z(t,x) &- \Delta_x Z(t,x) \\ &= -\int \frac{v}{N(t,x)} \left(\left(C - \left(1 + \frac{A}{2} \right) \right) + \frac{1}{2} (v - Bx)^2 + \int n(t,w) \, dw \right) n(t,x,v) \, dv \\ &+ C \int \frac{v}{N(t,x)} \left(\int \int \frac{n(t,x,v_*)n(t,x,v'_*)}{\int n(t,x,w) \, dw} Q(v,v_*,v'_*) \, dv_* \, dv'_* \right) \, dv \\ &- \left(1 - \frac{1}{2} (Z(t,x) - Bx)^2 - N + \frac{1}{2} \left(A - \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} \, dv \right) \right) N(t,x) \frac{Z(t,x)}{N(t,x)} \\ &+ 2 \partial_x (\log N(t,x)) \partial_x Z(t,x) \\ &= 2 \partial_x (\log N(t,x)) \partial_x Z(t,x) + (Bx - Z(t,x)) \int (v - Z(t,x))^2 \frac{n(t,x,v)}{N(t,x)} \, dv \\ &- \frac{1}{2} \int (v - Z)^3 \frac{n(t,x,v)}{N(t,x)} \, dv. \end{split}$$

Here we use the fact that the reproduction kernel does not affect the mean phenotypic trait: $\int vQ(v, v', v'_*) dv = \frac{v'+v'_*}{2}$.

6.4 Existence theory for the simplified model

We introduce W = Z - Dx in (18) to obtain the following equation

$$\partial_t W - \Delta_x W = -4 \frac{\partial_x W W}{1 - W^2} (\partial_x W + D) - W, \qquad (24)$$

with $-1 \leq W \leq 1$. This equation has a singularity for $W = \pm 1$.

The existence of singularities is an obstacle to have a well-defined problem. However, as we will see in Subsection 3.1, the singularities are fundamental to produce propagating fronts.

Similarly, we introduce $W_{\delta} := Z_{\delta} - Dx$. Then, (24) becomes

$$\partial_t W_{\delta} - \Delta_x W_{\delta} = -4 \frac{\partial_x W_{\delta} W_{\delta}}{1 - W_{\delta}^2 + \delta} (\partial_x W_{\delta} + D) - \frac{(1 - W_{\delta}^2) W_{\delta}}{1 - W_{\delta}^2 + \delta},\tag{25}$$

with

$$W_{\delta}(t=0,\cdot) = W_{\delta}^{0}(\cdot).$$

With this choice of approximation we avoid the singularities and transform the singularity in -1 into a stable steady state (the stability is for the ODE formulation presented in Subsection 3.1).

Under the assumption

$$-1 \le W_{\delta}^0 \le 1,\tag{26}$$

equation (25) has a smooth solution that stays between -1 and 1 by the maximum principle (see e.g. [2]). We prove a regularizing effect for equation (25) and we deduce that the W_{δ} 's converge to a viscosity solution of a variant of equation (24). Before stating the result, we first give the definition of viscosity solutions (see also [11, 2] for general introduction to this theory).

Definition 1 Viscosity solution: The function $u \in C([0, \infty) \times \mathbb{R})$ is a viscosity solution of the equation

$$L\left(\partial_{xx}^2 u, \partial_x u, \partial_t u, t, x\right) = 0, \qquad (t, x) \in (0, \infty) \times \mathbb{R},$$

if and only if for all $\phi \in C^{\infty}((0,\infty) \times \mathbb{R})$, we have

1. if $u - \phi$ has a local maximum at a point $(t_0, x_0) \in (0, \infty) \times \mathbb{R}$ then

$$L\left(\partial_{xx}^{2}\phi(t_{0},x_{0}),\partial_{x}\phi(t_{0},x_{0}),\partial_{t}\phi(t_{0},x_{0}),u(t_{0},x_{0}),t_{0},x_{0}\right) \leq 0.$$

2. if $u - \phi$ has a local minimum at a point $(t_0, x_0) \in (0, \infty) \times \mathbb{R}$ then

$$L\left(\partial_{xx}^{2}\phi(t_{0},x_{0}),\partial_{x}\phi(t_{0},x_{0}),\partial_{t}\phi(t_{0},x_{0}),u(t_{0},x_{0}),t_{0},x_{0}\right)\geq0.$$

Theorem 3 Under assumption (26), W_{δ} the solution of (25) verifies, for all $(t, x) \in \mathbb{R}_+ \times \mathbb{R}$,

$$-1 \le W_{\delta}(t,x) \le 1, \qquad |\partial_x W_{\delta}(t,x)| \le \min(L_2, \frac{1}{2\sqrt{t}} + L_1),$$
 (27)

with

$$L_{2} = \max\left(\sup_{x \in \mathbb{R}} \left|\partial_{x} W_{\delta}(x, 0)\right|, L_{1}\right) \in \mathbb{R}_{+} \cup +\infty$$

and L_1 a positive constant independent of δ . Moreover, after extraction of a subsequence, the W_{δ} 's converge to a continuous function W that is a viscosity solution of

$$(1 - W^{2})\partial_{t}W - (1 - W^{2})\Delta_{x}W = -4\partial_{x}WW(\partial_{x}W + D) - (1 - W^{2})W.$$
 (28)

We can easily verify that if the initial solution is as in Example 1, i.e. $W_{\delta}(t=0,\cdot) \equiv -1$, then we have $W_{\delta}(t,\cdot) \equiv -1$ for all t > 0. Therefore our approximation chooses the biological solution.

Proof of Theorem 3: We differentiate equation (24) with respect to x and obtain

$$\partial_t \partial_x W_{\delta} - \Delta_x \partial_x W_{\delta} = -4 \frac{W_{\delta}}{1 - W_{\delta}^2 + \delta} (2 \partial_x W_{\delta} + D) \partial_x (\partial_x W_{\delta}) -4 \partial_x W_{\delta}^2 (\partial_x W_{\delta} + D) \frac{1 + W_{\delta}^2 + \delta}{(1 - W_{\delta}^2 + \delta)^2} -\partial_x W_{\delta} \left(1 - \delta \frac{1 + W_{\delta}^2 + \delta}{(1 - W_{\delta}^2 + \delta)^2} \right),$$
(29)

where the last term comes from

$$\partial_x \left(\frac{W_\delta (1 - W_\delta^2)}{1 - W_\delta^2 + \delta} \right) = \partial_x \left(W_\delta - \frac{\delta W_\delta}{1 - W_\delta^2 + \delta} \right) = \partial_x W_\delta \left(1 - \delta \frac{1 - W_\delta^2 + \delta + 2W_\delta^2}{(1 - W_\delta^2 + \delta)^2} \right)$$

We multiply (29) by $\partial_x W_{\delta}$ and divide by $|\partial_x W_{\delta}|$ and obtain

$$\begin{aligned} \partial_t |\partial_x W_{\delta}| - \Delta_x |\partial_x W_{\delta}| &\leq -4 \frac{W_{\delta}}{1 - W_{\delta}^2 + \delta} (2\partial_x W_{\delta} + D) \partial_x (|\partial_x W_{\delta}|) \\ &- 4\partial_x W_{\delta}^2 (\partial_x W_{\delta} + D) \left(\frac{1 + W_{\delta}^2 + \delta}{(1 - W_{\delta}^2 + \delta)^2} \right) \operatorname{sgn}(\partial_x W_{\delta}) \\ &- \left(1 - \delta \frac{1 + W_{\delta}^2 + \delta}{(1 - W_{\delta}^2 + \delta)^2} \right) |\partial_x W_{\delta}|. \end{aligned}$$

It follows that, for $\delta < 1$,

$$\partial_t |\partial_x W_{\delta}| - \Delta_x |\partial_x W_{\delta}| \leq \alpha(t, x) \partial_x |\partial_x W_{\delta}| + \left(\frac{1+W_{\delta}^2 + \delta}{(1-W_{\delta}^2 + \delta)^2}\right) \cdot \left(-4|\partial_x W_{\delta}|^3 + 4D|\partial_x W_{\delta}|^2 + |\partial_x W_{\delta}|\right),$$

with

$$\alpha(t,x) = -4\frac{W_{\delta}}{1 - W_{\delta}^2 + \delta} \left(2\partial_x W_{\delta} + D\right).$$

Therefore for L_1 large enough and for $\delta < 1$, we have that $|\partial_x W_{\delta}|$ is a subsolution of the following equation

$$\partial_t g - \Delta_x g = \alpha(t, x) \partial_x g - 4 \left(\frac{1 + W_{\delta}^2 + \delta}{(1 - W_{\delta}^2 + \delta)^2} \right) (g - L_1)^3, \tag{30}$$

One can easily verify that, for $\delta < 1$, the functions $g_1(t,x) := L_2$ and $g_2(t,x) := \frac{1}{2\sqrt{t}} + L_1$ are supersolutions to equation (30) and they satisfy

$$|\partial_x W_{\delta}(0,x)| \le g_1(0,x), \qquad |\partial_x W_{\delta}(0,x)| \le g_2(0,x).$$

It follows that, for $\delta < 1$,

$$|\partial_x W_{\delta}(t,x)| \le \min(g_1(t,x), g_2(t,x)),$$

and thus (27).

Now we define

$$T_{\delta} := F(W_{\delta}) = (1+\delta)W_{\delta} - \frac{W_{\delta}^3}{3}.$$

We first consider the case with $L_2 < +\infty$. From (27), we deduce that the T_{δ} 's are uniformly bounded and Lipschitz in $[0, +\infty) \times \mathbb{R}$. Moreover we have

$$\partial_t T_{\delta} - \Delta T_{\delta} = (1 - W_{\delta}^2 + \delta) \partial_t W_{\delta} - (1 - W_{\delta}^2 + \delta) \Delta W_{\delta} + 2W_{\delta} |\nabla W_{\delta}|^2.$$

From the above equation, (25) and (27) we deduce that $\partial_t T_{\delta} - \Delta T_{\delta}$ is uniformly bounded in $[0, +\infty) \times \mathbb{R}$. It follows that the T_{δ} 's are uniformly continuous in time in $[0, +\infty) \times \mathbb{R}$ (see [3]). Using the Arzela Ascoli Theorem we conclude that, after extraction of a subsequence, the T_{δ} 's converge locally uniformly to a continuous function T in $[0, +\infty) \times \mathbb{R}$.

In the case with $L_2 = +\infty$, following the arguments above we obtain that the T_{δ} 's are uniformly bounded and locally uniformly Lipschitz in $(0, +\infty) \times \mathbb{R}$. The locally uniform convergence of the T_{δ} 's, along subsequences, is also hold in $(0, +\infty) \times \mathbb{R}$.

Finally from the fact that

$$F'(W_{\delta}) = 1 + \delta - W_{\delta}^2 > 0, \quad \text{for } -1 \le W_{\delta} \le 1,$$

we obtain that F is an invertible function. We write

$$W_{\delta} = F^{-1}(T_{\delta}),$$

with F^{-1} continuous. Therefore the W_{δ} 's converge locally uniformly to the continuous function $W = F^{-1}(T)$ along subsequences. The convergence takes place in $[0, +\infty) \times \mathbb{R}$, if $L_2 < +\infty$ and in $(0, +\infty) \times \mathbb{R}$ if $L_2 = +\infty$.

By the stability of viscosity solutions (see [11, 2]), we conclude that W is a viscosity solution of (28).

In Subsection 3.1 we study the propagating fronts for this model. To be able to compare the solutions with the propagating fronts and to show the propagation of the density in space, we need a comparison principle. Unfortunately as we saw above, the equation (28) does not have a unique viscosity solution and therefore it does not admit a comparison principle. However we can prove a comparison principle for the approached model. We first recall its definition:

Definition 2 Comparison principle: Equation $L(\partial_{xx}^2 u, \partial_x u, u, x, t) = 0$ verifies a comparison principle, if for any subsolution w^1 , that is

$$L(\partial_{xx}^2 w^1, \partial_x w^1, w^1, x, t) \le 0,$$

and supersolution w^2 of L, that is

$$L(\partial_{xx}^2 w^2, \partial_x w^2, w^2, x, t) \ge 0,$$

such that $w^1(0,x) \leq w^2(0,x)$, we have

$$w^1(t,x) \le w^2(t,x), \quad \text{for all } (t,x) \in \mathbb{R}_+ \times \mathbb{R}.$$

We prove that there is a comparison property for (25). In particular (25) has a unique solution. **Theorem 4** Equation (25) admits a comparison principle in the set of solutions $\{-1 \le W \le 1\}$. **Proof of Theorem 4:** We suppose that W_1 and W_2 are respectively subsolution and supersolution of (25) and

$$W_1(t=0,\cdot) \le W_2(t=0,\cdot).$$

We prove that $W_1 \leq W_2$ for all $(t, x) \in \mathbb{R}_+ \times \mathbb{R}$. Let (\bar{t}, \bar{x}) be a maximum point of $W_1 - W_2$. Since it is a maximum point we have $\partial_x W_1(\bar{t}, \bar{x}) = \partial_x W_2(\bar{t}, \bar{x}) = p$. Therefore we have

$$\begin{aligned} \partial_t (W_1 - W_2)(\bar{t}, \bar{x}) &- \Delta (W_1 - W_2)(\bar{t}, \bar{x}) \\ &\leq -4p \, (p+D) \frac{(1+W_1W_2 + \delta)}{(1-W_1^2 + \delta)(1-W_2^2 + \delta)} \, (W_1 - W_2)(\bar{t}, \bar{x}) \\ &- \frac{(1+\delta)(1-W_1^2 - W_1W_2 - W_2^2) + W_1W_2 + W_1^2W_2^2}{(1-W_1^2 + \delta)(1-W_2^2 + \delta)} \, (W_1 - W_2)(\bar{t}, \bar{x}). \end{aligned}$$

In the previous section we proved that $|\partial_x W|$ is bounded. Thus p(p+D) is bounded. Moreover W_1 and W_2 are bounded and

$$1 - W_i^2 + \delta \ge \delta$$
, for $i = 1, 2$.

Therefore the coefficient of $W_1 - W_2$ is bounded. Following the classical maximum principle we deduce that equation (25) admits a comparison principle.

6.5 Existence of propagating fronts

Theorem 5 For any D > 0, there exists $\nu_D \in \mathbb{R}$ such that (24) has a propagating front of speed ν , $Z(t, x) = Dx + U(x - \nu t)$, satisfying

$$U(x) \to 0 \text{ as } x \to -\infty, \quad U(x) \to -1 \text{ as } x \to +\infty,$$

if and only if $\nu > \nu_D$.

The propagating front with speed ν is unique (up to a translation), and ν_D is a decreasing function of D.

Theorem 6 The equation (18) has a non-trivial steady-state if and only if $\nu_D < 0$. Those steady-states Z(t,x) = Dx + U(x) satisfy

 $U(x) \to 1 \text{ as } x \to -\infty, \quad U(x) \to -1 \text{ as } x \to +\infty.$

To prove Theorems 5 and 6, we will use the two following Lemma:

Lemma 1 Let $\nu \in \mathbb{R}$. There exist only two, up to a shift in the t variable, solutions (u, v) to the ODE defined by the vector field (20) such that $(u, v)(t) \to (0, 0)$ as $t \to -\infty$. At most one of them is globally defined, which satisfies:

$$(u,v)(t) \sim -C_{-}e^{\frac{\sqrt{\nu^{2}+4}-\nu}{2}t}(1,\frac{\sqrt{\nu^{2}+4}-\nu}{2}).$$

Moreover, for this solution, u is strictly decreasing.



Figure 11: Solutions of the ODE defined by the vector field (20), on the left for D := 0.7 and $\nu = 0.5, 0.73141, 1$ (the vector field represented corresponds to $\nu = 0.73141$). A propagating front then exists for $\nu \ge 0.73141$ only, and the biologically relevant solution is obtained for $\nu = \nu_D \sim 0.73141$. On the right, we represent solutions of the ODE defined by the vector field (20), with $\nu = 0$ and D = 0.7.

Lemma 2 Let $\nu_1 \leq \nu_2$, and (u_{ν_1}, v_{ν_1}) , (u_{ν_1}, v_{ν_1}) be the corresponding solutions given by Lemma 1. If for $t_1, t_2 \in \mathbb{R}$,

$$\begin{cases} u_{\nu_1}(t_1) = u_{\nu_2}(t_2) \\ v_{\nu_1}(t_1) \le v_{\nu_2}(t_2), \end{cases}$$
(31)

then, for any $t'_1 > t_1$, $t'_2 > t_2$ such that $u_{\nu_1}(t'_1) = u_{\nu_2}(t'_2)$, we have $v_{\nu_1}(t'_1) \leq v_{\nu_2}(t'_2)$, and this inequality is strict if $\nu_1 < \nu_2$.

Proof of Lemma 1: The Differential of the vector field F in (0,0) is

$$DF_{(0,0)} = \left(\begin{array}{cc} 0 & 1\\ 1 & -\nu \end{array}\right)$$

Therefore (0,0) is a hyperbolic equilibrium point (det $DF_{(0,0)} < 0$). The Hartman-Grobman Theorem applies, and thus there exist only two (non-trivial) solutions (u, v) satisfying $(u, v)(t) \rightarrow_{t \rightarrow -\infty}$ (0,0). Since the eigenvector associated to the positive eigenvalue of $DF_{(0,0)}$ is $(1, \frac{\sqrt{\nu^2+4}-\nu}{2})$, the corresponding solutions are equivalent to

$$(u,v)(t) \sim_{t \to -\infty} \pm C_{\pm} e^{\frac{\sqrt{\nu^2 + 4} - \nu}{2}t} \left(1, \frac{\sqrt{\nu^2 + 4} - \nu}{2}\right),$$

for some $C_{-}, C_{+} > 0$.

The solution such that $(u, v)(t) \sim_{t \to -\infty} C_+ e^{\frac{\sqrt{\nu^2 + 4} - \nu}{2}t} \left(1, \frac{\sqrt{\nu^2 + 4} - \nu}{2}\right)$ satisfies $u(\bar{t}) > 0, v(\bar{t}) > 0$ for some \bar{t} . Since $F_U(u(\bar{t}), V) > 0$ for $V \ge v(\bar{t})$ and $F_V(U, v(\bar{t})) \ge F_V(u(\bar{t}), v(\bar{t})) > 0$ for $U \in [u(\bar{t}), 1)$, the solution cannot escape $[u(\bar{t}), 1) \times [v(\bar{t}), \infty)$. In particular, for $t \ge \bar{t}, u'(t) = v(t) \ge v(\bar{t}) > 0$ and since the vector field is not defined for U = 1, the solution cannot be global. The other solution satisfies $u(\bar{t}) < 0$, $v(\bar{t}) < 0$ for some \bar{t} arbitrarily small. Since $F_U(0, V) \leq 0$ for $V \leq 0$ and $F_V(U, 0) \leq 0$ for $U \leq 0$, the solution cannot escape \mathbb{R}^2_- . In particular, we have $u'(t) = v(t) \leq 0$, which shows that u is strictly decreasing with respect to t.

Proof of Lemma 2: Since u_{ν} is strictly decreasing, we can define the graph of (u_{ν_1}, v_{ν_1}) .

We assume that \bar{t}_1 , \bar{t}_2 are the smallest points respectively in (t_1, ∞) and in (t_2, ∞) such that $(u_{\nu_2}, v_{\nu_2})(\bar{t}_2) = (u_{\nu_1}, v_{\nu_1})(\bar{t}_1)$. We have,

$$\begin{aligned} v_{\nu_1}'(\bar{t}_1) &= -\nu_1 v_{\nu_1}(\bar{t}_1) + 4 \frac{u_{\nu_1}(t_1)v_{\nu_1}(t_1)}{1 - u_{\nu_1}(\bar{t}_1)^2} \left(v_{\nu_1}(\bar{t}_1) + D \right) + u_{\nu_1}(\bar{t}_1) \\ &= v_{\nu_2}'(\bar{t}_2) + (\nu_2 - \nu_1)v_{\nu_1}(\bar{t}_1) \\ &\leq v_{\nu_2}'(\bar{t}_2), \end{aligned}$$

this inequality being strict if $\nu_1 < \nu_2$. It follows that $\frac{v'_{\nu_1}(\bar{t}_1)}{u'_{\nu_1}(t_1)} > \frac{v'_{\nu_2}(\bar{t}_2)}{u'_{\nu_2}(t_2)}$. The graph of (u_{ν_2}, v_{ν_2}) can thus only cross the graph of (u_{ν_1}, v_{ν_1}) from below to above, when t increases. This is enough to conclude that $v_{\nu_1}(t'_1) \leq v_{\nu_2}(t'_2)$, for all $t'_1 > t_1$ and $t'_2 > t_2$. Moreover, the latter inequality is strict if $\nu_1 < \nu_2$. This completes the proof of Lemma 2.

Proof of Theorem 5: Step 1: We show that the solution (u, v) given by Lemma 1 satisfies either $v(\bar{t}) = -D$ for some \bar{t} , or $(u, v)(t) \to (-1, 0)$ as $t \to +\infty$. Moreover, the solution is global in this last case only.

Since u is strictly decreasing and $F_V(-1, V) = +\infty$ for $V \in (-D, 0)$, only two situations are possible: either $v(\bar{t}) = -D$ for some $\bar{t} < +\infty$, or $(u, v)(t) \to (-1, 0)$ as $t \to \bar{t} \in \mathbb{R} \cup \{+\infty\}$.

If $v(\bar{t}) = -D$, then v(t) < -D for all $t \ge \bar{t}$. This is because

$$F_V(U, -D) = \nu D + U \le F_V(u(\bar{t}), -D) \le 0$$

for $U \in (-1, u(\bar{t}))$ and since u is decreasing. Therefore, $u'(t) = v(t) \leq -D$ for all $t \geq \bar{t}$. From the latter, together with $F_V(-1, V) = -\infty$ for V < -D, we obtain that the solution cannot be globally defined.

Now let $(u, v)(t) \to (-1, 0)$ as $t \to \overline{t} \in \mathbb{R} \cup \{\infty\}$. We prove that $\overline{t} = \infty$. For (U, V) close to (-1, 0), we have $F_V(U, V) \sim -2D\frac{V}{1+U} - 1$. We deduce that,

$$\frac{d}{dt} \left(\frac{v}{1+u}\right) (t) = \frac{F_V(u(t), v(t))(1+u(t)) - v(t)F_U(u(t), v(t))}{(1+u(t))^2} \\ \sim \frac{1}{(1+u(t))^2} \left[\left(-2D\frac{v(t)}{1+u(t)} - 1\right) (1+u(t)) - v(t)^2 \right] \\ \sim \frac{1}{(1+u(t))^2} \left[-2Dv(t) - v(t)^2 - (1+u(t)) \right] \\ \geq 0,$$

if $v(t) \leq \frac{-1}{3D}(1+u(t))$ and (u(t), v(t)) is close to (-1, 0).

We assume that \tilde{t} is such that (u, v)(t) is close to (-1, 0) for $t \ge \tilde{t}$. We then have, $\frac{v(t)}{1+u(t)} \ge \min\left(\frac{v(\tilde{t})}{1+u(\tilde{t})}, \frac{-1}{3D}\right)$ for all $t \ge \tilde{t}$. It follows that $u'(t) = v(t) \ge -C(u(t) + 1)$, which implies the estimate

$$u(t) \ge -1 + (u(\tilde{t}) + 1)e^{-C(t-t)}.$$

Since $(u, v)(t) \to (-1, 0)$ as $t \to \overline{t}$, we deduce that $\overline{t} = +\infty$, and (u, v) is indeed globally defined.

Step 2: We next prove that there exists a constant ν_D such that, there exists a propagating front if and only if $\nu > \nu_D$.

If $\nu > \frac{1}{D}$, we have $F_V(U, -D) = \nu D - U > 0$ for $U \in (-1, 1)$. Therefore, the solution given by Lemma 1 cannot cross the line V = -D, and thus it defines a propagating front thanks to Step 1. We deduce that, there exists a propagating front if ν is large enough.

In $[-1/\sqrt{2}, 0] \times [-D, 0]$ and for $\nu \le -(4\sqrt{2}+1)D$, we have

$$F_V(U,V) \le -\nu V - 4\sqrt{2}V(V+D) \le DV \le DF_U(U,V).$$

It follows that the solution given by Lemma 1 necessarily crosses the line V = -D, and thus it does not define a propagating front according to Step 1. We deduce that the model does not admit a propagating front if $-\nu$ is large enough.

We consider a solution $(u_{\nu_1}, v_{\nu_1})(t)$ given by Lemma 1 for some ν_1 , that converges to (-1, 0) as $t \to +\infty$, and $\nu_2 > \nu_1$. Since $\frac{\sqrt{\nu^2+4}-\nu}{2}$ is a decreasing function of ν and

$$(u_{\nu_i}, v_{\nu_i})(t) \sim -C_- e^{\frac{\sqrt{\nu_{\nu_i}^2 + 4} - \nu_{\nu_i}}{2}} \left(1, \frac{\sqrt{\nu_{\nu_i}^2 + 4} - \nu_{\nu_i}}{2}\right),$$

the graph of $(u_{\nu_1}, v_{\nu_1})(t)$ is below the graph of $(u_{\nu_2}, v_{\nu_2})(t)$ for $t \ll 0$. Thanks to Lemma 2, this implies that the whole graph of (u_{ν_1}, v_{ν_1}) is below the graph of (u_{ν_2}, v_{ν_2}) . Using the latter and Step 1 we obtain that (u_{ν_2}, v_{ν_2}) defines a propagating front.

Finally, we show that ν_D is a decreasing function of D. Firstly we notice that, for $D_1 \leq D_2$, we have $F_U^{D_1} = F_U^{D_2}$, and $F_V^{D_1} \leq F_V^{D_2}$ in $(-1,0] \times \mathbb{R}_-$. We deduce that, using Step 1, if the solution given by Lemma 1 for D_1 converges to (-1,0) as $t \to \infty$, so does the one associated to D_2 . This proves that ν_D is a decreasing function of D.

Proof of Theorem 6: We assume that $\nu_D < 0$. Then, for $\nu = \frac{\nu_D}{2}$, the solution $(u_{\nu_D/2}, v_{\nu_D/2})$ given by Lemma 1, is globally defined, and it satisfies $(u_{\nu_D/2}, v_{\nu_D/2})(t) \rightarrow (-1, 0)$ as $t \rightarrow +\infty$. Moreover we have

$$(u_{\nu_D/2}, v_{\nu_D/2})(t) \sim_{t \to -\infty} -C_- e^{\frac{\sqrt{(\nu_D/2)^2 + 4} - \nu_D/2}{2}t} \left(1, \frac{\sqrt{(\nu_D/2)^2 + 4} - \nu_D/2}{2}\right),$$

and $\frac{\sqrt{(\nu_D/2)^2 + 4} - \nu_D/2}{2} > 1.$

Consider now the vector field (20) for $\nu = 0$. Since $(u_0, v_0)(t) \sim_{t \to -\infty} -C'_-e^t(1, 1)$, for \bar{t} small enough, $(u_0, v_0)(\bar{t})$ is strictly above the graph of $(u_{\nu_D/2}, v_{\nu_D/2})$. Let \tilde{v} be such that $(u_0(\bar{t}), \tilde{v})$ is strictly between $(u_0, v_0)(\bar{t})$ and the graph of $(u_{\nu_D/2}, v_{\nu_D/2})$. We define (\bar{u}, \bar{v}) to be the solution of the ODE given by the vector fields (20) such that $(\bar{u}, \bar{v})(0) = (u_0(\bar{t}), \tilde{v})$ and $\nu = 0$. Then, according to Lemma 2, (\bar{u}, \bar{v}) is defined in \mathbb{R}_+ . Moreover, since $(u_0, v_0)(t) \to (0, 0)$ as $t \to -\infty$ and (0, 0) is a hyperbolic equilibrium point, there exists $\tilde{t} > 0$ such that $\bar{u}(\tilde{t}) = 0$.

By symmetry, we have $(\bar{u}, \bar{v})(t+t) = (-\bar{u}, \bar{v})(t-t)$. Thus, (\bar{u}, \bar{v}) is globally defined and satisfies $(\bar{u}, \bar{v})(t) \to (\pm 1, 0)$ as $t \to \pm \infty$. This completes the proof of Theorem 6.

6.6 Approximation of the propagating fronts

In this subsection, we will study in more detail the propagating fronts of (18). As mentioned in Subsection 3.1, for each D > 0, there exists a one-parameter family of propagating fronts: for any $\nu \ge \nu_D$, there exists a unique propagating front of speed ν_D . By analogy to the KPP-Fisher equation (see e.g. [34, 46]), one can expect that there is only one stable propagating front, the one with the least speed $\nu = \nu_D$. In Section 4, numerical computations have confirmed the idea that the minimal speed ν_D is the natural speed to consider here.

For $\nu > \nu_D$, the propagating front $Dx + U_{\nu}$ is smooth (see Subsection 6.5), and corresponds to a solution (U, U') of (20) such that $(U, U')(x) \rightarrow_{x \to -\infty} (0, 0)$ and $(U, U')(x) \rightarrow_{x \to \infty} (-1, 0)$. The front with the minimal speed is slightly different: it is the uniform limit of the fronts $Dx + U_{\nu}$ as $\nu \rightarrow \nu_D$ (see Figure 11). The limit front $Dx + U_{\nu_D}$ has a singularity and cannot be a classical solution of (18). It is however a viscosity solution of (18). In the phase plane, this minimal speed front corresponds to a solution $(U, V)|_{\mathbb{R}_-}$ of (20), defined on \mathbb{R}_- , such that $(U, V)(x) \rightarrow_{x \to -\infty} (0, 0), (U, V)(x) \rightarrow_{x \to 0^-} (-1, -D)$, and $Dx + U_{\nu_D}(x) = Dx - 1$ for x > 0.

Our aim is to approach the particular solutions (U, V) of (20) that joint (0, 0) to (-1, -D). To do this, we introduce the ansatz that for $x \in (-\infty, 0]$,

$$V = \alpha U + \beta |U|^{3/2}.$$
(32)

To estimate α and β , we proceed as follows:

- Since (U, V)(0) = (-1, -D), necessarily, $\alpha = D + \beta$.
- Since (0,0) is a saddle point of the vector field (20), (U,V) must be tangent to the eigenvector $(1, (\sqrt{\nu^2 + 4} \nu)/2)$ close to (0,0) (see Subsection 6.5 in the appendix). Then,

$$\beta = \frac{1}{2} \left(\sqrt{\nu_D^2 + 4} - \nu_D \right) - D.$$
(33)

• Finally, close to (-1, -D),

$$\left\{ \begin{array}{l} F_U(U,V) \sim -D \\ F_V(U,V) \sim \nu_D D + 2D \frac{V+D}{U+1} - 1 \end{array} \right.$$

 $\frac{\partial V}{\partial U}(x=0)$ should then satisfy

$$\frac{\partial V}{\partial U}(x=0) = \frac{-1}{D} \left(\nu_D D + 2D \frac{\partial V}{\partial U}(x=0) - 1 \right),$$

and our ansatz yields $\frac{\partial V}{\partial U}(x=0) = D - \beta/2.$

If we combine the estimates above, we obtain

$$0 = \frac{1}{D} - \frac{9D}{2} - \frac{7}{4}\nu_D + \frac{3}{4}\sqrt{\nu_D^2 + 4},$$

and thus, ν_D satisfies:

$$\nu_D \sim \frac{1}{20} \left(7 \left(\frac{2}{D} - 9D \right) + 3 \sqrt{\left(\frac{2}{D} - 9D \right)^2 + 40} \right).$$

This approximation of ν_D is not rigorous, but it is however very close to the numerical estimations of ν_D , as shown in Figure 3. This speed can be positive (the population then invades the empty territory), or negative (the population disappears).

We can go further and give an approximation of the propagating front with minimal speed, which satisfies

$$U'(x) = (D+\beta)U(x) + \beta |U(x)|^{3/2},$$
(34)

for $x \in \mathbb{R}_{-}$, and U(x) = -1 for x > 0. We recall that β is given by (33), with ν_D estimated by (21). To solve (34), we use the change of variable $U(x) = -\phi^{-2}$. Replacing this in (34) we obtain that $\phi' = -\frac{D+\beta}{2} + 2\beta$ on \mathbb{R}_{-} , and thus $\phi(x) = Ce^{-\frac{D+\beta}{2}x} + \frac{\beta}{D+\beta}$. We can then compute U, recalling that U(0) = -1:

$$U(x) = -\left(\left(1 - \frac{\beta}{D+\beta}\right)e^{-\frac{D+\beta}{2}x} + \frac{\beta}{D+\beta}\right)^{-2}, \quad x \in \mathbb{R}_{-},$$

and U(x) = -1 on \mathbb{R}_+ .

References

- KE Atkins, JML Travis, Local adaptation and the evolution of species' range under climate change. J. Theoret. Biol., 266, 449–457 (2010).
- [2] G Barles, Solutions de viscosité des équations de Hamilton-Jacobi. Mathématiques & Applications(Belin) vol. 17, Springer-Verlag, Paris, 1994.
- [3] G Barles, S Biton, O Ley, A geometrical approach to the study of unbounded solutions of quasilinear parabolic equations. Arch. Rational Mech. Anal 162, 287–325 (2002).
- [4] NH Barton, Adaptation at the edge of a species range. 365–392 in J. Silvertown and J. Antonovics, eds. Integrating ecology and evolution in a spatial context. Vol. 14. Blackwell, Oxford (2001).
- [5] E Bouin, V Calvez, N Meunier, S Mirrahimi, B Perthame, G Raoul, R Voituriez, Invasion fronts with variable motility: phenotype selection, spatial sorting and wave acceleration. Preprint, *submitted*.
- [6] MG Bulmer, The mathematical theory of quantitative genetics. Oxford, UK: Clarendon Press (1980).

- [7] MG Bulmer, Maintenance of genetic variability by mutation-selection balance: a child's guide through the jungle. *Genome*, **31**(2), 761–767 (1989).
- [8] R Bürger, The Mathematical theory of selection, recombination and mutation. *Wiley*, New-York (2000).
- [9] TJ Case, ML Taper, Interspecific Competition, Environmental Gradients, Gene Flow, and the Coevolution of Species' Borders. Am. Nat. 155(5), 583–605 (2000).
- [10] A Calsina, S Cuadrado, L Desvillettes, G Raoul, Asymptotics of steady states of a selection mutation equation for small mutation rate, Preprint, *submitted*.
- [11] M. G. Crandall, H. Ishii, and P.-L. Lions, User's guide to viscosity solutions of second order partial differential equations. Bull. Amer. Math. Soc. (N.S.) 27(1), 1–67 (1992).
- [12] L Desvillettes, PE Jabin, S Mischler, G Raoul, On selection dynamics for continuous populations. *Commun Math Sci* 6(3), 729–747 (2008).
- [13] O Diekmann, PE Jabin, S Mischler, B Perthame, The dynamics of adaptation: an illuminating example and a Hamilton-Jacobi approach. *Theor. Popul. Biol.* 67(4), 257–71 (2005).
- [14] M Doebeli, HJ Blok, O Leimar, U Dieckmann, Multimodal pattern formation in phenotype distributions of sexual populations. emphProc Biol Sci. 274(1608), 347–57 (2007).
- [15] M Doebeli, U Dieckmann, Speciation along environmental gradients. Nature 421, 259–264 (2003).
- [16] RA Fisher, The wave of advance of advantageous genes. Ann. Eugenics 7, 353–369 (1937).
- [17] O Hallatschek, DR Nelson, Life at the front of an expanding population. Evolution 64(1), 193–206 (2010).
- [18] A Hastings, K Cuddington, KF Davies, CJ Dugaw, S Elmendorf, A Freestone, S Harrison, M Holland, J Lambrinos, U Malvadkar, BA Melbourne, K Moore, C Taylor, D Thomson, The spatial spread of invasions: new developments in theory and evidence. *Ecol. lett.* 8(1), 91–101 (2005).
- [19] JK Hill, CD Thomas, DS Blakeley, Evolution of flight morphology in a butterfly that has recently expanded its geographic range. *Oecologia* 121, 165–170 (1999).
- [20] JK Hill, CD Thomas, B Hutley, Climate and habitat availability determine 20th century changes in a butterfly's range margin. Proc. R. Soc. Lond. B 266, 1197–1206 (1999).
- [21] P Hinow, F Le Foll, P Magal, GF Webb, Analysis of a model for transfer phenomena in biological populations. SIAM J. Appl. Math. 70, 40–62 (2009).
- [22] BB Huey, GW Gilchrist, ML Carlson, D Berrigan, L Serra. Rapid Evolution of a Geographic Cline in Size in an Introduced Fly. *Science* 287(5451), 308–309 (2000).
- [23] JE Keymer, P Galajda, C Muldoon, S Park, RH Austin, Bacterial metapopulations in nanofabricated landscapes. Proc. Natl. Acad. Sci. U.S.A 103, 17290–17295 (2006).

- [24] M Kimura. A stochastic model concerning the maintenance of genetic variability in quantitative characters. Proc. Natl. Acad. Sci. U.S.A 54, 731–736 (1965).
- [25] M Kirkpatrick, NH Barton, Evolution of a species' range, Amer Nat 150(1), 1–23 (1997).
- [26] A Kremer, O Ronce, JJ Robledo-Arnuncio, F Guillaume, G Bohrer, R Nathan, JR Bridle, R Gomulkiewicz, EK Klein, K Ritland, A Kuparinen, S Gerber, S Schueler, Long-distance gene flow and adaptation of forest trees to rapid climate change. *Ecol. lett.* **15**, 378–392 (2012).
- [27] CE Lee, Evolutionary genetics of invasive species. Trends Ecol. Evol. 17(8), 386–391 (2002).
- [28] A Lorz, S Mirrahimi, B Perthame, Dirac mass dynamics in a multidimensional nonlocal parabolic equation, Comm. Partial Differential Equations 36(6) 1071–1098 (2011).
- [29] P Magal, G Raoul, in preparation.
- [30] G Martin, SF Elena, T Lenormand, Distributions of epistasis in microbes fit predictions from a fitness landscape model. *Nature Genetics* **39** 555–560 (2007).
- [31] E Mayr, Animal species and evolution. Harvard University Press, Belknap, Cambridge, Mass (1963).
- [32] JAJ Metz, R Nisbet, SAH Geritz, How should we define fitness for general ecological scenarios ? Trends Ecol. Evol. 7, 198–202 (1992).
- [33] S Mirrahimi, G Raoul, Asexual populations structured by a space variable and a phenotypical trait. *in progress*.
- [34] J.D. Murray, Mathematical Biology II: Spatial models and biomedial applications. Third Edition, Springer (2003).
- [35] B Neuffer, H Hurka, Colonization history and introduction dynamics of Capsella bursapastoris (Brassicaceae) in North America: isozymes and quantitative traits. *Mol. Ecol.* 8, 1667–1681 (1999).
- [36] CP Pease, R Lande, JJ Bull, A model of population growth, dispersal and evolution in a changing environment. *Ecology* 70, 1657–1664 (1989).
- [37] JR Peck, JM Yearsley, D Waxman, Explaining the geographic distributions of sexual and asexual populations. *nature* **391**, 889–892 (1998).
- [38] BL Phillips, GP Brown, RS Webb, R Shine, Invasion and the evolution of speed in toads. *Nature* 439, 803–806 (2006).
- [39] BL Phillips, GP Brown, RS Webb, R Shine, Reid?s Paradox Revisited: The Evolution of Dispersal Kernels during Range Expansion. Am. Nat. 172, S34–S48 (2008).
- [40] J Polechova, N Barton, G Marion, Species' range: Adaptation in Space and Time. Amer Nat 174(5), 186–204 (2009).

- [41] C Prevost, Applications of partial differential equations and their numerical simulations of population dynamics. *PhD Thesis*, University of Orleans (2004).
- [42] O Ronce, M Kirkpatrick, When sources become sinks: migrational meltdown in heterogeneous habitats. *Evolution* 55, 1520–1531 (2001).
- [43] S Roy, JP"Simon, FJ Lapointe, Determination of the origin of the cold-adapted populations of barnyard grass (Echinochloa crus-galli) in eastern North America: a total-evidence approach using RAPD DNA and DNA sequences. *Can. J. Bot.* **12**, 1505–1513 (2000).
- [44] DB Saakian, A New Method for the Solution of Models of Biological Evolution: Derivation of Exact Steady-State Distributions. J. Stat. Phys. 128(3), 781–798 (2007).
- [45] AD Simmons, CD Thomas, Changes in dispersal during species' range expansions. Am. Nat. 164(3), 378–395 (2004).
- [46] N Shigesada, K Kawasaki. Biological inveasions: theory and practice. Oxford series in Ecology and Evolution, Oxford University Press (1997).
- [47] CD Thomas, EJ Bodsworth, RJ Wilson, AD Simmons, ZG Davies, M Musche, L Conradt, Ecological and evolutionary processes at expanding range margins. *Nature* 411, 577–581 (2001).
- [48] JMJ Travis, K Mustin, TG Benton, C Dytham, Accelerating invasion rates result from the evolution of density-dependent dispersal. J. Theoret. Biol. 259, 151–158 (2009).
- [49] J Tufto, Quantitative genetic models for the balance between migration and stabilizing selection. Genet. Res. Camb. 76, 285–293 (2000).
- [50] M Turelli, N H Barton, Genetic and statistical analyses of strong selection on polygenic traits: what, me normal? *Genetics* 138, 913–941 (1994).
- [51] C Villani, A review of mathematical topics in collisional kinetic theory. In Handbook of mathematical fluid dynamics, 1, North-Holland, Amsterdam, 71–305 (2002).
- [52] C. Villani, Topics in Optimal Transportation, volume 58 of Graduate Studies in Mathematics. American Mathematical Society (2003).
- [53] D Waxman, JR Peck. The frequency of the perfect genotype in a population subject to pleiotropic mutation, *Theor Popul Biol* **69**, 409–418 (2006).